Author’s response to reviews

Title: Risk Factors Associated with Poor Health Outcomes for Children under the Age of 5 with Moderate Acute Malnutrition in Rural Fagita Lekoma District, Awi Zone, Amhara, Ethiopia, 2016

Authors:

Dube Jara (jaradube@yahoo.com)

W/Amilak Adamu (woldeamlakadamu@gmail.com)

Mulunesh Alemayehu (mulu2019@gmail.com)

Sahai Burrowes (sahai.burrowes@tu.edu)

Version: 2 Date: 05 Sep 2017

Author’s response to reviews:

We made all the revision requested like paper organization and language.

Through our revision we made radical change in the particularly in the language revision.

Thank you for your constructive comments